Author’s response to reviews

Title: The Diabetes Care Project: an Australian multicentre, cluster randomised controlled trial (study protocol)

Authors:

Matthew J Leach Dr (matthew.leach@unisa.edu.au)
Leonie Segal Prof (leonie.segal@unisa.edu.au)
Adrian Esterman Prof (adrian.esterman@unisa.edu.au)
Caroline Armour Ms (caroline.armour86@gmail.com)
Robyn McDermott Prof (robyn.mcdermott@unisa.edu.au)
Tim Fountaine Dr (Tim_Fountaine@mckinsey.com)

Version: 2 Date: 30 May 2013

Author’s response to reviews:

To whom it may concern:

The following information is a response to the query of peer review for the Diabetes Care Project.

Prior to securing funding for the project from the Australian Department of Health and Ageing, a peer review process was conducted by experts from many diabetes organisations nationally. These organisations include the Department of Health and Ageing, McKinsey and Company, South Australia Department of Health, Queensland Department of Health, General Practice Queensland together with Brisbane South division of general practice, Ipswich division of general practice, Wide Bay Medicare Local and Darling Downs South West Queensland Medicare Local, Healthfirst Network, Adelaide Hills division of general practice, the University of South Australia, Baker IDI, Statewide Diabetes Clinical Network Queensland, Diabetes Australia, Diabetes Australia Queensland and Health Consumers Alliance of South Australia.

Kind regards,
Dr Tim Fountaine